Author's response to reviews

Title: Early onset lung cancer, cigarette smoking and the SNP309 of the murine double minute-2 (MDM2) gene

Authors:

Kirstin K Mittelstrass (kirstin.mittelstrass@gsf.de)
Wiebke W Sauter (wiebke.sauter@gsf.de)
Albert A Rosenberger (arosenb@gwdg.de)
Thomas T Illig (illig@gsf.de)
Maria M Timofeeva (m.timofeeva@dkfz-heidelberg.de)
Norman N Klopp (klopp@gsf.de)
Hendrik H Dienemann (hendrik.dienemann@thoraxklinik-heidelberg.de)
Eckart E Meese (hgemee@med-rz.uni-sb.de)
Gerhard G Sybrecht (prof.g.w.sybrecht@uniklinikum-saarland.de)
Gabi G Woelke (woelke@gsf.de)
Mathias M Cebulla (mathias.cebulla@sanktgeorg.de)
Maria M Degen (maria.degen@t-online.de)
Harald H Morr (harald.morr.waldhof@t-online.de)
Peter P Drings (peter.drings@thoraxklinik-heidelberg.de)
Andreas A Groeschel (inagro@med-rz.uni-sb.de)
Karsten K Grosse Kreymborg (karsten.g.kreymborg@medizin.uni-leipzig.de)
Karl K Haeussinger (gauting@asklepios.com)
Gerd G Hoeffken (hoeffken@fachkrankenhaus-coswig.de)
Christine C Schmidt (schmidt@fachkrankenhaus-coswig.de)
Bettina B Jilge (b.jilge@skc.de)
Wilhelm W Schmidt (e.schmidt@skc.de)
You-Dschun Y Ko (y.ko@jk-bonn.de)
Dagmar D Taeuscher (d.taeuscher@t-online.de)
Jenny J Chang-Claude (j.chang-claude@dkfz-heidelberg.de)
Heinz-Erich H-E Wichmann (wichmann@gsf.de)
Heike H Bickeboeller (hbickeb@gwdg.de)
Angela A Risch (a.risch@dkfz-heidelberg.de)

Version: 4 Date: 21 April 2008

Author's response to reviews: see over
“Early onset lung cancer, cigarette smoking and the genetic polymorphism SNP309 of the murine double minute-2 (MDM2) gene” (BMC-Cancer/2007/1996041118160143)

Dear BioMed Central Editorial Production Team,

Thank you for publishing our manuscript in your journal. We have itemized our changes below in a point by point fashion and have revised the manuscript accordingly. All authors have read the revised paper.

Thank you very much for your efforts.

Sincerely Yours,

Kirstin Mittelstraße

# Description of changes:

**Authors list:** The list of authors in the manuscript is now exactly written as they are in the submission system.

**Manuscript section:** The manuscript now includes all sections asked for in the right order. The Abbreviations are now behind the Conclusions.

**Table order:** The tables are now provided in the text in the same order then they are provided at the end of the manuscript. We have therefore changed the position of table 2 and 3.